# UC San Diego

UC San Diego Previously Published Works

## Title

Modeling epistasis in mice and yeast using the proportion of two or more distinct genetic backgrounds: Evidence for "polygenic epistasis"

Permalink https://escholarship.org/uc/item/1gh2b8b7

Journal PLOS Genetics, 16(10)

ISSN

1553-7390

Authors

Rau, Christoph D Gonzales, Natalia M Bloom, Joshua S <u>et al.</u>

Publication Date

2020

DOI 10.1371/journal.pgen.1009165

Peer reviewed



# 

**Citation:** Rau CD, Gonzales NM, Bloom JS, Park D, Ayroles J, Palmer AA, et al. (2020) Modeling epistasis in mice and yeast using the proportion of two or more distinct genetic backgrounds: Evidence for "polygenic epistasis". PLoS Genet 16(10): e1009165. https://doi.org/10.1371/journal. pgen.1009165

Editor: David A. Buchner, Case Western Reserve University School of Medicine, UNITED STATES

Received: May 21, 2020

Accepted: October 2, 2020

Published: October 26, 2020

**Copyright:** © 2020 Rau et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data from the HMDP may be accessed at https://systems. genetics.ucla.edu/ All data from the AlL cross may be accessed at http://palmerlab.org/protocols-data/ All data from the Yeast crosses may be accessed at https://github.com/joshsbloom/yeast-16-parents The GxO algorithm may be found at https://github. com/ChristophRau/GxTheta.

**Funding:** This work was supported by: National Institute on Drug Abuse(AAP: R01DA021336;

RESEARCH ARTICLE

# Modeling epistasis in mice and yeast using the proportion of two or more distinct genetic backgrounds: Evidence for "polygenic epistasis"

Christoph D. Rau<sup>1</sup>, Natalia M. Gonzales<sup>2</sup>, Joshua S. Bloom<sup>3</sup>, Danny Park<sup>4</sup>, Julien Ayroles<sup>5</sup>, Abraham A. Palmer<sup>6</sup>, Aldons J. Lusis<sup>3</sup>, Noah Zaitlen<sup>7</sup>\*

 Department of Genetics, University of North Carolina at Chapel Hill, Chapel Hill, NC, United States of America, 2 Department of Human Genetics, University of Chicago, Chicago, IL, United States of America,
Department of Human Genetics, UCLA, Los Angeles, CA, United States of America, 4 Department of Medicine, UCSF, San Francisco, CA, United States of America, 5 Department of Ecology and Evolutionary Biology, Princeton University, Princeton, NJ, United States of America, 6 Department of Psychiatry, and Institute for Genomic Medicine, UCSD, San Diego, CA, United States of America, 7 Department of Neurology, UCLA, Los Angeles, CA, United States of America

\* nzaitlen@mednet.ucla.edu

# Abstract

## Background

The majority of quantitative genetic models used to map complex traits assume that alleles have similar effects across all individuals. Significant evidence suggests, however, that epistatic interactions modulate the impact of many alleles. Nevertheless, identifying epistatic interactions remains computationally and statistically challenging. In this work, we address some of these challenges by developing a statistical test for *polygenic epistasis* that determines whether the effect of an allele is altered by the global genetic ancestry proportion from distinct progenitors.

#### **Results**

We applied our method to data from mice and yeast. For the mice, we observed 49 significant genotype-by-ancestry interaction associations across 14 phenotypes as well as over 1,400 Bonferroni-corrected genotype-by-ancestry interaction associations for mouse gene expression data. For the yeast, we observed 92 significant genotype-by-ancestry interactions across 38 phenotypes. Given this evidence of epistasis, we test for and observe evidence of rapid selection pressure on ancestry specific polymorphisms within one of the cohorts, consistent with epistatic selection.

### Conclusions

Unlike our prior work in human populations, we observe widespread evidence of ancestrymodified SNP effects, perhaps reflecting the greater divergence present in crosses using mice and yeast. NMG: F31DA036358) (www.drugabuse.gov) National Institute of General Medical Sciences (NMG: T32GM007197, JFA: R35GM124881) (www.nigms.nih.gov) National Institutes of Health (CDR: K99HL138301, NZ: K25HL121295, U01HG009080, and R01HG006399 AJL: U54DK120342, R01HL147883, R01 DK117850) (www.nih.gov) The funding bodies played no role in the study other than funding.

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

Many statistical tests which link genetic markers in the genome to differences in traits rely on the assumption that the same polymorphism will have identical effects in different individuals. However, there is substantial evidence indicating that this is not the case. Epistasis is the phenomenon in which multiple polymorphisms interact with one another to amplify or negate each other's effects on a trait. We hypothesized that individual SNP effects could be changed in a polygenic manner, such that the proportion of as genetic ancestry, rather than specific markers, might be used to capture epistatic interactions. Motivated by this possibility, we develop a new statistical test that allowed us to examine the genome to identify polymorphisms which have different effects depending on the ancestral makeup of each individual. We use our test in two different populations of inbred mice and a yeast panel and demonstrate that these sorts of variable effect polymorphisms exist in 14 different physical traits in mice and 38 phenotypes in yeast as well as in murine gene expression. We use the term "polygenic epistasis" to distinguish these interactions from the more conventional two- or multi-locus interactions.

## Introduction

Genetic association studies in humans and model organisms have identified a number of significant links between individual polymorphisms and phenotypic variability. A fundamental assumption of many of these studies is that an allele will have a similar effect in each member of the population, that is, that epistatic and other higher-order interactions across the genome can largely be ignored[1–4]. Prior studies to specifically detect epistasis in flies[5], yeast[6], mice[7,8], and humans[9] have considered pairwise interactions between loci to begin to identify interacting loci, yet this approach is hampered by a substantial increase in the threshold of significance due to the increased numbers of tests (from n to at least  $\binom{n}{2}$ ), where n are the number of SNPs tested). In this study, we consider detection of epistasis by testing the effects of interactions between individual polymorphisms and global ancestry ( $\theta$ ), which we defined as the percentage of the genome inherited from a given ancestor or ancestral population.

Recently, we developed a SNP-ancestry interaction model and applied it to a human dataset, detecting only modest evidence of ancestry-specific genetic effects[10]. There is substantial evidence of interactions with ancestry in model organisms. For example, we observed radically different phenotypic consequences of null alleles of *Tcf7l2* and *Cacna1c* when expressed on different inbred strain backgrounds[11]. Recently, a number of powerful genetic resources have been developed for model systems to map variation in complex traits, such as the Hybrid Mouse Diversity Panel[12], the Collaborative Cross[13,14], the BXD recombinant inbred panel[15], various outbred mouse and rat populations[16], or the Drosophila Synthetic Population Resource[17]. The variation in ancestry present in these populations provide a unique opportunity to test for ancestry-driven epistasis, but no existing method is suitable for this task.

To address this problem, we incorporate a linear mixed model capable of accounting for the highly structured relatedness of constructed model organism[18] into our previous approach. We then examine two distinct mouse cohorts as well as a panel of yeast crosses and test for interactions between the effects of individual polymorphisms and the percentage of the genome for each individual that originated with a specific founder. Although our initial model was designed to identify both genetic background and environmental effects, the controlled

environment in which model organisms are raised is intended to minimize sources of environmental variance. This test, which we have called  $Gx\theta$ , acts as a surrogate measure of the concurrent action of many other SNPs, and allows us to ask whether or not the effect size of a given SNP changes as a function of overall genetic ancestry. In contrast to a pairwise epistatic test, where significance means a detected interaction between two specific loci, a natural interpretation of a significant  $Gx\theta$  interaction is widespread epistasis of the tested genotype G with many loci across the genome, as we previously showed[10].

We first evaluated the statistical properties of our method on simulated phenotypes based on real recombinant inbred line data sets[12]. The Gx $\theta$  test successfully distinguishes between the marginal effects driven only by a given SNP or ancestry and effects that arise from their interaction. We then applied the test to two different populations of mice: recombinant inbred (RI) lines that are a subset of the HMDP[12] (143 phenotypes), and a 50<sup>th</sup> generation intercross between the inbred LG/J and SM/J mouse strains[19], whose ancestry proportions have been clearly and precisely determined[19] (75 phenotypes). To show that our approach is broadly applicable, we also apply this test to a yeast panel consisting of thousands of progeny derived from 15 distinct yeast crosses[20] (40 phenotypes). Utilizing gene expression data from the HMDP cohort in two different conditions, we further demonstrate that our approach replicates findings across similar populations.

Given this evidence of epistasis, we examined the ancestry distribution at specific sites in the mouse HMDP RI lines. Under neutrality, on average, we would expect equal representation of founder ancestry in the panel, however, we observed that for many positions in the genome there is a statistically significant depletion or overrepresentation of specific founder ancestries. We interpret this as evidence of selection during the process of RI strain derivation and we identify regions of the genome with strong selection up to 4 Mb in length. These regions are evidence of strong Gx $\theta$  interactions which inhibit or promote the transmission of specific alleles to subsequent generations. These regions are enriched for genes involved in cancer and organogenesis and are enriched (P = 1.8E-4) for metallopeptidases, which play key roles in fertility and neo/perinatal lethality[21–23].

Our observation of genomic locations where the effect size of a given polymorphism is driven by overall genomic context and ancestry highlights the importance of studying epistasis and other effects in model organisms. Studying these context-dependent effects in human cohorts has been notoriously difficult, but our results indicate that given the right design and analytical framework these effects can be highly significant, and has implications for our ability to build predictive models from genotype to phenotype and account for ancestry effects in population studies.

#### Results

#### Method overview

Our objective is to determine whether a given SNP i has a different effect size as a function of changes in overall genetic background. As described in our previous work[10], this is effectively achieved through an interaction test between each SNP and global genetic ancestry  $\theta$ . In a standard test of the association between a polymorphism  $X_i$  and a phenotype y, we can model this relationship as  $y = \mu + \beta_i X_i + u + e$ , where  $\mu$  is the mean phenotypic value,  $\beta$  is the effect size of the SNP, u is a random variable accounting for relatedness and e is a random variable accounting for all other sources of error in the study. We extend this analysis to account for interactions between a polymorphism and ancestry by adding terms to our model such that the relationship now becomes

$$\mathbf{y} = \mu + \beta_i \mathbf{X}_i + \delta \boldsymbol{\theta} + \beta_{G_{\mathbf{X}} \theta_i} \boldsymbol{\theta} * \mathbf{X}_i + \mathbf{u} + \mathbf{e}$$

where  $\delta$  is the global weight of the ancestry effect,  $\theta$  are the ancestries for all N individuals and  $\beta_{Gx\theta}$  are the weights of the Gx $\theta$  effect. Our Gx $\theta$  test is then an LRT test with a null of  $\beta_{Gx\theta i} = 0$  and an alternate of  $\beta_{Gx\theta i} \neq 0$ .

To account for multiple testing, we follow the best practices of the respective model organism communities, specifically we use the previously reported genome-wide significance threshold for the HMDP of 4.2E-6 [24] or the AIL of 8.06E-6 [19]. For Yeast, we follow the example previously reported of Bloom et al [20] and use an FDR of 5%.

#### Simulated data

To examine the properties of our approach in a model organism cross, we applied the method to phenotypes simulated using real genotypes that reflect the underlying relatedness present within the recombinant inbred (RI) lines of the Hybrid Mouse Diversity Panel (HMDP). Random SNPs with a minor allele frequency (MAF) between 25% and 75% were selected from our panel and phenotypes constructed at 200 values of either a main SNP effect  $\beta_G$  or a SNP-ancestry interaction effect  $\beta_{Gx\theta}$  where  $\theta$  is the proportion of all SNPs arising from C57BL/6J (see Methods) mice for interaction with each individual SNP with 1,000 simulations at each  $\beta$  (200,000 total simulations) (see Methods).

We observe that changes to either  $\beta_G$  or  $\beta_{Gx\theta}$  independently did not affect the power of identifying an association for  $\beta_{Gx\theta}$  or  $\beta_G$ , respectively, indicating that these two terms are correctly being estimated independently of one another. We briefly explored incorporating a second genetic relationship matrix (GRM) accounting for population structure arising from descent from a single ancestor (the B6 mouse)[25], however we observed no significant improvement in power when incorporating this second GRM (Fig 1A and 1B).

To evaluate what forms of interactions the test is powered to identify we examined a range of epistatic architectures, focusing specifically on our method's ability to recover higher-order interactions, a type of epistasis which traditional models struggle to accommodate. For 100,000 simulations, a SNP was selected at random from SNPs with MAF between 25 and 75% and combined with 1–10 additional SNPs (10,000 each) and phenotypes generated for these SNP groupings followed by analysis by the Gx $\theta$  algorithm (see Methods). As the number of interacting SNPs increases, the power to detect the individual SNP decreases as expected (Fig 1C), however the power to detect a Gx $\theta$  locus rises (Fig 1D), suggesting that our model is powerfully situated to identify SNPs involved in complex higher-order interactions, especially those consisting of more than 5 interacting partners (S1 Fig).

#### Application to in vivo data

**Mouse populations.** We applied the Gx $\theta$  test to two large panels of mice to identify Gx $\theta$  effects (i.e. instances where a given polymorphism interacts epistatically with one or many other loci as captured in the model by  $\theta$ , the global ancestry). Our first cohort is the HMDP, a set of 150 + commercially available inbred strains[26]. Numerous GWAS have been performed in the HMDP, including several using PYLMM, which forms the core of our algorithm[24,27,28]. The largest component of the HMDP is comprised of 122 RI strains (28 AxB, 71 BxD, 12 BxH, 11 CxB). Each RI was constructed from re-derivation of novel inbred lines via brother-sister mating following an F<sub>2</sub> cross between the sub-panel's parental lines. In the case of the HMDP, one of the parental strains for each RI strain was the commonly studied strain C57BL/6J (**B6**). Using B6 as the common ancestral line to the RI panels, we calculated  $\theta$ s for each RI strain (Fig 2A). As expected, the average ancestry attributable to B6 was roughly 50% (50.63%) and roughly normally distributed. We removed a single outlier, BXD32/TyJ, whose B6 Ancestry of 25.41% reflects a previously known additional backcross to DBA/2J, resulting in a strain that is 75% DBA and 25% B6.



**Fig 1. Simulation results** A+B) Power calculations based on simulated data with variable main SNP effects  $\beta_G$  (A) or variable Ancestry-SNP effects  $\beta_{Gx\theta}$  (B). Blue and Purple power curves are the power curves for detecting a significant SNP effect, while the orange and green curves are the power curves for detecting a significant Gx $\theta$  effect. Two phenotypic models, one incorporating 1 GRM (1K) correcting for relatedness in the SNPs (green, purple) and one incorporating 2 GRMs (2K) correcting for relatedness in both SNPs and Ancestry (red, blue) were used. (C+D) Effect of increasing numbers of epistatically interacting SNPs on the algorithms ability to detect a significant main SNP effect (C) or Gx $\theta$  effect (D). The straight line in C and D represents the line where expected and observed P values are the same, while the grey band indicates the 95% confidence interval in which the observed P-value distribution cannot be distinguished from the expected P-value distribution.

https://doi.org/10.1371/journal.pgen.1009165.g001

Each study using the HMDP uses a different subset of the entire panel, and we selected a study on heart failure induced by the chronic beta adrenergic agonist isoproterenol[29] for analysis as it used the most RI strains compared to other published HMDP data. We used 123 clinical pheno-types in conjunction with microarray-derived gene expressions measured in the left ventricle and on average tested 67 RI strains per phenotype.





https://doi.org/10.1371/journal.pgen.1009165.g002

Cohort	Phenotype	Chromosome	Basepair	rsID	Р
AIL	Act3.2	11	93607101	rs27076868	6.22E-07
AIL	Average Body Weight	6	133867165	rs30057768	1.62E-06
AIL	Glucose	9	70318603	rs50320206	3.03E-07
AIL	Weight at 68 days	6	133899245	rs47230944	2.09E-06
HMDP	E wave 3 weeks PT	19	21360156	rs36743940	5.71E-10
HMDP	E/A ratio 3 weeks PT	6	8462274	rs47261338	7.10E-08
HMDP	Fibrosis control	17	41256115	rs6295287	6.74E-07
HMDP	Fractional shortening control	X	48286693	rs30272504	3.69E-06
HMDP	Free fatty acids control	11	113035483	rs27020574	4.09E-06
HMDP	Heart rate 1 weeks PT	18	9408717	rs29769121	2.37E-06
HMDP	Heart rate control	X	48286693	rs30272504	2.68E-06
HMDP	Left atrial weight 3 weeks PT	16	54550810	rs51319671	2.18E-06
HMDP	LVIDs control	X	48286693	rs30272504	2.92E-06
HMDP	Left ventricular mass 1 weeks PT	3	148140470	rs31313229	2.91E-06
HMDP	Left ventricular mass 2 weeks PT	17	72564961	rs50549031	8.65E-08
HMDP	Right atrial weight 3 weeks PT	16	49837267	rs50479702	4.13E-07
Yeast	Cross 2999: EGTA	XIV	622639	NA	2.91E-07
Yeast	Cross 2999: Lithium Chloride	IV	519675	NA	1.04E-07
Yeast	Cross 2999: Lithium Chloride	XII	341678	NA	1.16E-05
Yeast	Cross 2999: Manganese Sulfate	XIV	622639	NA	2.30E-06
Yeast	Cross 3043: Manganese Sulfate	XVI	206053	NA	4.17E-10
Yeast	Cross 3043: Paraquat	XV	410166	NA	3.14E-08
Yeast	Cross 3043: YNB	XV	561457	NA	3.67E-07
Yeast	Cross 3043: YPD	XV	561457	NA	2.50E-06
Yeast	Cross B: Maltose	VII	1067754	NA	1.20E-07

Table 1. Most significant Gxt associations for each phenotype observed in the AIL or HMDP cohorts as well as all Yeast Gxt loci passing a cross-wide 5% FDR	
threshold See S1 Table for complete list of loci. Act3.2 is 'activity levels in control animal on day 3 of a conditioned place preference test', LVIDs is 'left ventricular into	er-
nal dimension at systole' and PT is post isoproterenol treatment.	

https://doi.org/10.1371/journal.pgen.1009165.t001

The second cohort consists of 1,063 animals from the  $F_{50}$  – $F_{56}$  generation of an advanced intercross line (AIL) created by crossing the LG/J and SM/J inbred mouse strains [19]. Unlike the RI strains from the HMDP, AILs are maintained in a manner that minimizes inbreeding and promotes genetic diversity at each generation. We arbitrarily set LG/J as the ancestral strain of interest and calculated  $\theta$ s for each of the 1,063 mice in the panel (Fig 2B). For this study, we focused on a diverse set of 133 phenotypes that had been measured in these mice. We describe the results of the Gx $\theta$  associations in each panel before demonstrating replication of interactions in a phenotypic trait as well as in expression data.

**Evidence of SNP x ancestry interactions in phenotypic data.** We applied the  $Gx\theta$  method to the 123 observed heart-failure related phenotypes from the HMDP RI strains. We observed well-calibrated statistics, with genomic inflation factors  $\lambda_{GC}$  equal to 0.978 and  $\lambda_{Gx\thetaC}$  equal to 1.045 after pooling the p values across all phenotypes. We observed 44 significant  $Gx\theta$  loci across 9 phenotypes: E/A ratio, free fatty acid content in the blood, cardiac fibrosis, fractional shortening of the heart during contraction, heart rate, internal diameter of the left ventricle, left ventricular mass and left and right atrial weights (Tables 1 and S1). These  $Gx\theta$  loci were largely distinct from previously reported GWAS loci in the same phenotypes in this panel of mice[27,30], yet contained a number of highly relevant genes, as discussed below.

By way of example, we focus on two important phenotypes from the HMDP panel. Cardiac fibrosis is a marker of cardiac dysfunction. Genes identified through the  $Gx\theta$  screen (Fig 3B)



**Fig 3. Gx0 results in the yeast, HMDP and LGxSM AIL Cohorts.** A) Lithium Chloride Exposure in Yeast B) Cardiac Fibrosis and C) Left Ventricular Weight in the HMDP D) Glucose in the AIL. Significance Threshold in Yeast (A) = 1.2E-5 (FDR 5%) HMDP (B,C) = 4.2E-6[24], AIL (D) = 9.1E-6 (FDR 5%). Summary Statistics may be found in S6–S9 Tables.

https://doi.org/10.1371/journal.pgen.1009165.g003

as potential candidates for cardiac fibrosis include: *Crisp2* (rs6295287, p = 6.74E-7), a secreted biomarker of cardiovascular disease[31], *Top2b* (rs31538570, p = 2.14E-06), which plays a cardioprotective role in response to stress[32], *Rarb* (rs31538570, p = 2.14E-06), a known regulator of inflammation with unknown function in the heart[33], and *Fibrosin* (rs33146511, p = 2.54E-6), a major component of the fibrosis pathway[34]. An increase in left ventricular mass in response to catecholamine challenge is the primary marker of cardiac hypertrophy in the HMDP. A single Gx $\theta$  locus(rs31313229, p = 2.91E-06) on chromosome 3 (Fig 3C) contains the gene *Lphn2*, which has a role in the promotion of cellular adhesion in response to external stimuli.[35] Crucially, none of these loci, many of which contain candidates of particular interest or relevance to heart failure, were reported in the original study, suggesting the possibility of new targets for further therapeutic research.

We next applied the Gx $\theta$  method to 133 phenotypes measured in the LG/J x SM/J AIL, including the sex of the animal as a covariate as, unlike the HMDP data, the AIL consists of both male and female animals. As with the HMDP data, we observed well-calibrated statistics in this cross, with genomic inflation factors  $\lambda_{GC}$  equal to 0.995 and  $\lambda_{Gx\theta C}$  equal to 1.033. Despite the larger sample size, we observed only 4 significant loci across 4 phenotypes: activity levels in a saline-injected animal on day 3 of a conditioned place preference test; average weight of animals across 5 different time points roughly a week apart; glucose (mg/dL) in blood after a 4 hour fast (Fig 3D); weight at ~68 days of age (Tables 1 and S1).

**Yeast populations.** We also used the Gx $\theta$  algorithm to examine 15 previously reported yeast crosses[20] ranging from 650 to 950 progeny per cross for evidence of significant Gx $\theta$  loci and an average of 71,000 segregating genetic variants. We examined the responses of these progeny to forty different chemical stimuli[20] and recovered 92 loci across 38 phenotypes which met a 5% FDR threshold for that cross/phenotype combination (S2 Table). At a more stringent level of significance (5% FDR across an entire cross), we identify 9 loci across 7 phenotypes (LiCl, EGTA, Maltose, MnSO4, Paraquat, YNB, YPD) which show strong evidence of Gx $\theta$  interactions (Table 1, Fig 3A).

#### Gxθ associations in HMDP cohort gene expression

We next examined gene expression in the hearts of the HMDP cohort using previously obtained transcriptome microarrays from both control and treated conditions[36]. Each cohort consists of approximately 70 RI lines, with 66 lines overlapping between the two

cohorts (full lists of strains in S2 Table). We examined 13,155 expressed and varying (Coefficient of Variation > 5%) genes from the left ventricles of the HMDP cohorts using the ~170k SNPs with MAF > = 0.05. We observed 1,486 significant associations with 18 genes at a Bonferroni-corrected P value of 3.2E-10 (135,130 associations with 1,350 genes at GW-significant threshold of 4.2E-6) in the control cohort and 597 significant associations with 39 genes at the same threshold in the treated cohort (32,043 associations with 1,042 genes at the genome-wide significant threshold of 4.2E-6[24]) (Full results available in S1 Table).

Phenotypic variability was larger in the catecholamine treated cohort of the heart failure HMDP study (average Coefficient of Variation in control was 22.6% vs 28.6% for the treated cohort)[29], which could partially explain the lower number of significant associations observed in that group of mice. Notably, we saw no enrichment for SNPs in the *cis*-regions of the examined genes in contrast to what is observed in standard mapping. Genes in the control cohort with significant Gx $\theta$  associations are enriched for mitochondria-associated genes (P = 1.8E-10 (P = .032 in treated)), suggesting a role of Gx $\theta$  interactions in the regulation of mitochondrial dynamics. Genes in the treated cohort with significant severe enriched for genes involved in post-transcriptional modifications to RNA including RNA splicing (P = 7.4E-3), highlighting the importance of alternative splicing to the response to catecholamine challenge[37].

**Replication of Gx0** Associations across eQTLs from HMDP Cohort. To demonstrate that the method is able to replicate Gx0 results across cohorts, we examined the reproducibility of expression Gx0 QTLs in the treated and untreated RI lines of the HMDP. Of the 1,486 associations observed in the control data, we observe 305 (21%) with Gx0 interaction (P < .05) in the treated cohort (36 at FDR< 5%), indicating a strong replication (p = 2.2E-16 from a binomial test) between the two cohorts despite differences in genetic background and the potentially reduced power in the treated cohort caused by the effects of the catecholamine drug. All significant associations may be found in S10 Table

#### Distortion from expected allelic frequencies

After population admixture, allelic frequencies will shift in the resulting population. This can be due to the effects of selection against deleterious allelic combinations (i.e. fitness epistasis), meiotic drive (i.e. intragenomic conflict resulting in the transmission of one or more alleles over another during meiosis). Alternatively, allele frequencies may shift through random genetic drift. We hypothesize that  $Gx\theta$  loci can be driven by fitness epistasis, or instances where allelic variation in one locus affects fitness at other loci across the genome. This hypothesis is in line with prior research[38–40] which has explored allelic imbalance or gametic ratio distortion as a function of epistasis and fitness. Examining the genomes of admixed population for signs of significant allelic shift may reveal interesting loci that can inform long standing questions in evolutionary and population genetics about the forces that maintains variation in fitness in populations, as well as the presence of non-equal chromosomal inheritance.

We therefore conducted a test to search for individual loci with enriched or depleted B6 ancestry in the RI strains. As expected, the average B6 ancestry across all SNPs was 50.68% +/-8.1%, which is indistinguishable from the ancestry by strain average of 50.63% +/- 6.7% (Fig 4A). At the level of individual loci, however, we observed significant variation in B6 ancestry across the genome (Fig 4B), with some SNPs displaying very low or very high frequency (Fig 4C). We calculated the statistical likelihood of detecting the observed ancestries at all loci across the genome ( $\lambda = 1.07$ , Fig 4D).

We subsequently performed a Wald test for ancestry frequency ( $\theta$ ) at each SNP i where  $W = \frac{(\bar{\theta} - \theta_i)^2}{var(\theta)}$ . We used the global variance in the denominator, which assumes that most SNPs



Fig 4. Allelic frequency imbalances of individual sites across the genome. A) A histogram of B6 ancestry across the geneome. B) Genome-wide plot of ancestry C) Zoomed in region on chr 12 with significant depletion of B6 allele. D) QQ plot of all Pvalues calculated by a Wald test. Genomic Inflation Factor  $\lambda = 1.07$ .

https://doi.org/10.1371/journal.pgen.1009165.g004

lie under the null and which is conservative if this assumption is violated and could be replaced with a more powerful test. For each SNP, we examined strains which were not invariant for that locus and identified 614 SNPs from the 170k original SNPs with significantly altered ancestries at an FDR of 5% (S3 Table). We suspect that these SNPs do not represent passive effects such as genetic drift due to the nature of how RI panels are created, with a single F2 generation of mice followed by brother-sister mating and eventual fixation as similar effects on allelic balance have been attributed to genetic incompatibilities in other mouse populations [41]. Allelic balance of paternal alleles across the RI panel, therefore, may be modeled using a Bernoulli distribution. In the HMDP, with 122 RI lines, the odds of a 30% imbalance (80% B6 vs 20% other) would be 8.3E-12, which, even across 200,000 SNPs results in 2E-6 expected SNPs with this level of imbalance across the entire genome.

Nine loci were identified in which 5 or more SNPs were located together (Table 2). These 9 regions contain the majority (525, 86%) of all significant affected loci and indicate regions where significant allelic imbalances are occurring. One region of particular interest lies on chromosome 18 and contains 246 SNPs (40% of all significantly altered SNPs) with an average B6 ancestry of 82.5%. SNPs in this region are found between B6 and C3H/HeJ and B6 and

82.50

0.017

<i>δ δ 1</i>									
Chromosome	Start	End	Number of SNPs	Average B6 Ancestry	Average Significance	Genes of Interest			
1	10763382	10945149	21	10.66	7.96E-05	Сраб			
1	59458339	60046978	54	16.22	5.50E-04	Bmpr2			
1	134463208	134712156	10	87.52	0.0048	Nfasc			
11	64495476	64538963	5	18.18	0.0011	Myocd			
11	110824338	110984075	18	85.75	0.0077	Kcnj2			
12	79002481	80362465	72	18.43	0.0011	Gphn			
12	106889886	107633641	49	19.15	0.0014	Pigh			
15	4743169	6121864	50	18.14	0.0011	Prkaa1			
	1	1	1	1	1	Î.			

246

Table 2. Regions with significant and sustained B6 depletion or selection.

https://doi.org/10.1371/journal.pgen.1009165.t002

5465201

\_\_\_\_\_

18

8965173

Epc1

BALB/cJ. This locus contains a number of interesting candidate genes potentially relating to organismal fitness, including *Epc1*, a transcription factor linked to DNA repair, muscle differentiation and cancer suppression[42,43]. It is also the only gene within the locus whose expression has a suggestive Gx $\theta$  association with the locus itself (P = 0.014).

When all genes within 2 MB (the average Linkage Disequilibrium (LD) block size in the HMDP[26]) of boundaries of these allelic imbalance loci are examined as a whole, we observe significant enrichments for genes involved in cancer (corrected P-values range from 1.3E-8 to 1.6E-7). These genes, whose functions range from cell cycle control to organismal growth and DNA damage mediation and repair are not unexpected given their crucial role in organismal survival. If the B6 or alternative allele deleteriously affect the function of one or more of these genes when combined with new alleles from other strains located elsewhere on the genome, then it is reasonable to assume that the ability for an individual to pass on its genetic information to the next generation may be significantly reduced. Other enriched gene ontology categories include the crucial-for-life categories of respiratory tube development (corrected pvalue = 2.7E-4) and heart development (corrected p-value = 7.6E-4), likely mediated through changes in Matrix Metalloproteinase activity (corrected p-value = 3.9E-8). These enzymes are canonically responsible for the regulation of the extracellular matrix, but have been linked to the modulation of responses to many bioactive molecules and have important roles in fertility, embryonic development and neo/perinatal mortality in addition to many other diseases[23]. Taken together, these enrichments suggest that we may be observing depletion due to epistatic interactions with other loci which were neutral or beneficial in the original inbred lines, but which become deleterious for a given allele when placed into a new genetic environment.

#### Discussion

We present a test which we call  $Gx\theta$ , that leverages admixed populations such as inbred mouse strains to identify sites of "polygenic epistasis" where a SNP interacts with an unknown number of other loci summarized into a single genomic ancestry score,  $\theta$ . One major advantage of this approach is that, unlike other epistasis-focused association approaches, it does not increase the number of tests when compared to a typical GWAS, so similar significance thresholds may be used.

The existence of epistatic interactions and their role in human diseases and phenotypes have been known for many years, with notable examples of gene-gene interactions in Alzheimer's disease[44], Bardet-Biedl syndrome[45] as well as classic interactions governing hair color and skin pigmentation. Despite these examples and many more like them, epistatic interactions have proven difficult to find using association techniques[46]. One notable exception has been the success of using consomic strains of mice, where one or more chromosome in strain A is replaced with a chromosome from another strain B[47]. This approach, which increases or decreases the contribution of a given strain to phenotypic and genetic variance in a clear and controlled manner has resulted in the identification of numerous epistatic interactions[48,49].

Our method sought to circumvent this traditional pitfall in Epistasis GWAS by examining each SNP only once for an interaction with a global genomic ancestry to identify interactions or "polygenic epistasis". Our model is able to identify much higher order interactions than can be observed using traditional SNP-SNP interaction models. As demonstrated through simulation, our model in fact increases in power as the complexity of the epistatic interactions increases, likely reaching peak power in the hypothetical scenario where every SNP interacts with the tested SNP to affect phenotypic expression. When applied to human populations, a different form of our test was able to find several significant associations[10]

In this study, we examined two populations of mice: the recombinant inbred lines of the Hybrid Mouse Diversity Panel[26] and an Advanced Intercross Lines (AIL) based on LG/J and

SM/J[19] as well as a yeast RIL mapping panel founded by fifteen yeast strains[20]. We observed significant Gx $\theta$  interactions in 14 of the 15 yeast crosses that affected over half of the examined phenotypes, reinforcing the widespread role of genotype-by-genotype interactions in regulating quantitative traits across diverse genetic backgrounds and phenotypes. In the mouse cohorts, however, despite nearly ten times the number of genetically distinct mice as well as a larger number of phenotypes and genotypes in the AIL, we observed approximately ten times more significant Gx $\theta$  peaks in the HMDP phenotypes than in the AIL phenotypes (44 vs 4).

Several reasons may account for this difference. First, outbred mice have significantly lower power due to increased phenotypic variance caused by high rates of heterozygosity at alleles while inbred populations have increased power due to a lack of this heterozygosity. This can also be seen in our regular GWAS results [19,29,30]. Second, the different phenotypes studied in each cohort will have different genetic architecture and experimental noise (e.g. the AIL study includes many behavioral traits while the HMDP does not). Third, significantly more recombination events have occurred in the AIL compared to the HMDP, leading to a tighter distribution of background and a larger number of independent tests. Fourth, the genetic backgrounds of the two cohorts are different, which might also contribute to differences in observed numbers of  $Gx\theta$ interactions. Although individually, LG and SM are only slightly less genetically diverged from one another as any pair of strains that make up the RI panels of the HMDP (S4 Table), the presence of five ancestral lines in the HMDP compared to only two in the F50 cross results in much more genetic diversity in the HMDP compared to the F50 mice. Finally, we observe a higher variance in ancestral background in the mice HMDP when compared to the AIL. As our method relies on differences in ancestral background to identify sites with different effect sizes in different genetic contexts, the reduced variance in the AIL lines necessarily corresponds to a decrease in the power to detect  $Gx\theta$  interactions. Taken together, our method is best suited to datasets with relatively low heterozygosity, clear and numerous differences in genetic background, and with higher variance in the percentage of SNPs attributable to a given ancestry.

Furthermore, we identify nine regions across the genome of the RI lines of the HMDP which show a strong selection for or against a particular parental allele. We propose that this is a result of an epistatic fitness interaction, where individual loci in one ancestral strain are genetically incompatible with alleles from another strain. We considered other alternatives explanations, some of which were explored in prior work[40]. One alternative would be population structure, where recombination may not have had sufficient time to randomly assort alleles after a bottleneck event, though we think that this is an unlikely explanation for our results. One of the key benefits of working with synthetic mapping populations such as RI panels is that the crossing scheme (e.g. fennel cross, round-robin) is designed to minimize such effects[39]. Another possibility would be assortative mating, where mate selection drives the selection for or depletion of a particular allele. While possible, and not explicitly arising from epistasis per se, it would still produce a fitness differential across allelic combinations and be picked up by our analysis. Finally, epistatic interactions, the focus of this manuscript, can lead to changes in fitness as demonstrated by prior work<sup>[40]</sup> which examined Drosophila RI panels in which all expected genotypic combinations for a diallelic cross were generated. They demonstrated a strong correlation between the degree of allelic imbalance and the effect size of the fitness effect of each allelic combination. It follows that similar scenarios should exist in mouse and yeast panels as well. One possible extension of this approach would be to study fitness in the Collaborative Cross strains, another animal population in which a large proportion of the strains went extinct, presumably due to genetic incompatibilities.

In the future, our approach could be leveraged to study additional extant populations or newly created cohorts. For example, our tool could be used to examine whether genetic modifications such as knockouts have differential effects across the mice of a cross, and we are actively exploring its utility in this regard.

#### Conclusion

The results of our study suggest that heterogenous SNP effects due to differing ancestries is pervasive in model organism populations. This observation is consistent with prior observations of epistatic interactions in inbred strains of mice as well as examples of ancestry interaction in human studies[10]. However, the number and magnitude of ancestry interactions we found was much larger than those found in human studies; we hypothesize this is driven by the larger genetic distance of the ancestors. Further analyses of the sites of polygenic epistasis may reveal novel epistatic interactions which drive phenotypic expression, and suggests that careful attention to genetic ancestry should be considered when studying the role of an individual polymorphism on a phenotype.

#### Methods

#### Model organism populations and ancestry calling

Model organism data were drawn from previously reported studies[19,20,29]. The LGxSM AIL consists of 1,063 G50-56 mice derived from an original F1 intercross between the LG and SM inbred lines. The Hybrid Mouse Diversity Panel consists of over 150 strains of commercially available inbred mice[26], of which 122 strains were recombinant inbred lines and suitable for our study. The yeast crosses originated from 15 original strains which were bred in pairs to create 15 new crosses. SNPwise ancestries were determined by identifying all SNPs which differed between parental lines (AIL: LG and SM or HMDP: C57BL/6 and A, C3H, DBA/2 or BALB/c or Yeast: See <u>S5 Table</u>). Genotypes from the G50-56.RI lines or yeast crosses were filtered for these SNPs and ancestries calculated using either LG or C57BL/6 or one of the yeast strains as the strain background of interest.

#### Simulation framework

**Power calculations.** We created sets of simulated phenotypes based on the genotypes of the HDMP RI panel, which is an admixed population in which the B6 strain, on average, contributes 50% of each strain's DNA. For each simulated phenotypes, we drew a SNP (75% > MAF > 25%) at random from the HMDP genotypes and created a phenotype based on  $\beta$ , the genetic effect size,  $\phi$  the effect size of the interaction between global ancestry ( $\theta$ ) and the chosen SNP and a multivariate normal (mvn) derived from three variance terms:  $\sigma_g^2$ , the proportion of variance attributable to genetic effects  $\sigma_{\theta}^2$ , the proportion of variance attributable to Gx $\theta$  effects and  $\sigma_{e}^2$ , the residual proportional variance attributable to all combined sources of error and variation not considered in this study. Phenotypes were generated both with and without the Gx $\theta$  variance term to ascertain the necessity of incorporating a second GRM ( $\mathbf{K}^A$ ), which would correct for relatedness in ancestry, into the algorithm.

We simulated four distinct phenotypes for our analysis

1. Phenotypes generated by including a SNP Effect

$$y = \beta X + mvn(0, \sigma_s^2 \mathbf{K} + \sigma_e^2 \mathbf{I}) \text{ or}$$
$$y = \beta X + mvn(0, \sigma_s^2 \mathbf{K} + \sigma_e^2 \mathbf{K}^A + \sigma_e^2 \mathbf{I})$$

2. Phenotypes generated by including a  $Gx\theta$  Effect

$$y = \beta_{Gx\theta}\theta * X + mvn(0, \sigma_g^2 \mathbf{K} + \sigma_e^2 \mathbf{I}) \text{ or}$$
$$y = \beta_{Gx\theta}\theta * X + mvn(0, \sigma_g^2 \mathbf{K} + \sigma_\theta^2 \mathbf{K}^A + \sigma_e^2 \mathbf{I})$$

In each phenotype,  $\sigma_g^2$  was set to 0.4. When incorporated,  $\sigma_\theta^2$  was set to 0.2 and  $\sigma_\varepsilon^2$  was set to the remainder of the variance (0.6 or 0.4). The power of our model and independence of our  $\beta$  and  $\beta_{Gx\theta}$  terms were queried by varying either  $\beta$  or  $\beta_{Gx\theta}$  from 0 to 1 (200 values set 0.005 apart) with 1,000 simulated phenotypes at each step (200,000 total simulations per phenotype).

**Epistasis simulation.** Using the same panel and set of SNPs described above, we drew one SNP at random to be our test SNP and one to ten additional SNPs to be simulated epistatically interacting SNPs (10,000 simulations per set of interacting SNPs, 100,000 simulations in total). For each simulation we created a composite SNP in which only strains with the minor (non-B6) allele in every one of the tests and interacting SNPs had that allele in the composite SNP. We used this composite SNP to generate a phenotype as described above with a  $\chi^2$  test statistic for the 2–11 interacting SNPs set to a large value of 20 to ensure a consistent and observable effect that could be recovered using either regular GWAS with pyLMM or our approach.

#### Gxθ model

The equation to determine the effects of a SNP and a SNP x Ancestry term on a phenotype can be written as:

$$y_{k} = \mu + \sum_{i=1}^{M} \beta_{i} X_{i} + \delta \theta_{k} + \sum_{i=1}^{M} \beta_{G x \theta_{i}} \theta_{i} X_{ik} + \varepsilon_{k}$$

Where  $y_k$  is the phenotype of individual k,  $\mu$  is the mean phenotypic value, M is the number of markers,  $\beta$  are the weights on the SNPs, X is the *m* by *n* array of SNP genotypes,  $\delta$  is the global weight of the ancestry effect,  $\theta$  is the ancestries for all N individuals.  $\beta_{Gx\theta}$  are the weights of the Gx $\theta$  effect and  $\varepsilon$  is the combined error term. We want to identify SNPs where  $\beta_{Gx\theta i} \neq 0$  as these are sites where Ancestry is interacting with our genotypes.

Motivated by our model above, we can write a new model for the effect of a single SNP i on a phenotypic trait as:

$$\mathbf{y} = \boldsymbol{\mu} + \beta_i \mathbf{X}_i + \delta \boldsymbol{\theta} + \beta_{G_{\mathcal{X}} \theta_i} \boldsymbol{\theta} * \mathbf{X}_i + \boldsymbol{u} + \boldsymbol{e}$$

For an individual SNP i. Here,  $\theta$  is the column vector of ancestries, and  $\theta^* X$  is the elementwise product. The random effect u accounts for relatedness of individuals based on SNPs. Our Gx $\theta$  test is then an LRT test with a null of  $\beta_{Gx\theta_i} = 0$  and an alternate of  $\beta_{Gx\theta_i} \neq 0$ .

#### Reporting $Gx\theta$ results and identifying genes of interest

GWAS thresholds were previously determined to be 4.1E-6 in the case of the HMDP[24] or 8.06E-6 in the case of the AIL[19]. An FDR of 5% was used in the yeast crosses as previously reported[20] Genes of Interest were identified by examining nearby genes (within 2 MB) for cis-eQTLs as previously reported in the HMDP[29] and/or by literature and database analyses using bioGPS (www.biogps.org) which draws data from PubMed and other publicly available databases to help researchers understand the role of their genes of interest.

#### Determining significance of allelic imbalance sites

Allelic imbalance significance was determined by performing a Wald test for ancestry frequency ( $\theta$ ) at each SNP i where  $W = \frac{(\bar{\theta} - \theta_i)^2}{var(\theta)}$  and only strains where the SNP was not invariant were included. Allelic imbalances were considered to be significant after Benjamini-Hochberg correction at a cutoff of 5%.

#### Gene set enrichment analysis

All gene enrichments were performed using the GeneAnalytics platform[50], which uses a binomial distribution to assess the enrichment for a user-supplied list of genes within GO terms, Superpaths and other biologically relevant categories. Reported P-values have been corrected for multiple comparisons using the Benjamini-Hochberg correction.

#### Declarations

Ethics approval and consent to participate: Not Applicable Consent for publication: Not Applicable

#### Supporting information

S1 Fig. Genomic Inflation Factors of SNP and Gxθ associations with increasingly complex SNP-SNP interactions.

(TIF)

S1 Table. All observed Gxθ associations between clinical traits and SNPs in the AIL, HMDP and Yeast cohorts. (XLSX)

S2 Table. Strains contained in the recombinant inbred panels of the HMDP and their presence in the control or isoproterenol-treated cohorts of the data. (XLSX)

**S3** Table. Alleles with significantly altered ancestries at FDR 5% in the HMDP. (CSV)

S4 Table. Number of SNPs between strains in the AIL and HMDP founder strains as determined by the mouse diversity array. Homozygous differences are 1, heterozygous differences are 0.5.

(XLSX)

**S5** Table. Information on the yeast crosses used in this manuscript. (XLSX)

S6 Table. Summary statistics for  $Gx\theta$  results from Fig 3A. (CSV)

S7 Table. Summary statistics for  $Gx\theta$  results from Fig 3B. (CSV)

**S8 Table. Summary statistics for Gxθ results from Fig 3C**. (CSV)

**S9** Table. Summary statistics for Gxθ results from Fig 3D. (CSV)

**S10** Table. All significant Gxθ associations for HMDP transcriptome data. (CSV)

#### Acknowledgments

The authors would like to thank Yibin Wang for his constructive critiques of the algorithm and the manuscript.

#### **Author Contributions**

Conceptualization: Christoph D. Rau, Danny Park, Noah Zaitlen.

Data curation: Christoph D. Rau, Natalia M. Gonzales, Joshua S. Bloom.

Formal analysis: Christoph D. Rau.

Funding acquisition: Christoph D. Rau, Noah Zaitlen.

Investigation: Christoph D. Rau, Natalia M. Gonzales, Joshua S. Bloom.

Methodology: Christoph D. Rau, Danny Park, Julien Ayroles, Noah Zaitlen.

Project administration: Noah Zaitlen.

**Resources:** Christoph D. Rau, Natalia M. Gonzales, Joshua S. Bloom, Abraham A. Palmer, Aldons J. Lusis.

Software: Christoph D. Rau.

Supervision: Noah Zaitlen.

Validation: Christoph D. Rau.

Visualization: Christoph D. Rau.

Writing - original draft: Christoph D. Rau, Noah Zaitlen.

Writing – review & editing: Christoph D. Rau, Natalia M. Gonzales, Joshua S. Bloom, Danny Park, Julien Ayroles, Abraham A. Palmer, Aldons J. Lusis, Noah Zaitlen.

#### References

- 1. Polderman TJC, Benyamin B, de Leeuw C a, Sullivan PF, van Bochoven A, Visscher PM, et al. Metaanalysis of the heritability of human traits based on fifty years of twin studies. Nat Genet. Nature Publishing Group; 2015; 47:702–709. https://doi.org/10.1038/ng.3285 PMID: 25985137
- Hill WG, Goddard ME, Visscher PM. Data and theory point to mainly additive genetic variance for complex traits. PLoS Genet. Public Library of Science; 2008; 4:e1000008. https://doi.org/10.1371/journal. pgen.1000008 PMID: 18454194
- Fish AE, Capra JA, Bush WS. Are Interactions between cis-Regulatory Variants Evidence for Biological Epistasis or Statistical Artifacts? Am J Hum Genet. Elsevier; 2016; 99:817–830. https://doi.org/10.1016/ j.ajhg.2016.07.022 PMID: 27640306
- Mäki-Tanila A, Hill WG. Influence of gene interaction on complex trait variation with multilocus models. Genetics. Genetics Society of America; 2014; 198:355–67. https://doi.org/10.1534/genetics.114. 165282 PMID: 24990992
- Huang W, Richards S, Carbone MA, Zhu D, Anholt RRH, Ayroles JF, et al. Epistasis dominates the genetic architecture of Drosophila quantitative traits. Proc Natl Acad Sci U S A. 2012; https://doi.org/10. 1073/pnas.1213423109 PMID: 22949659
- Costanzo M, VanderSluis B, Koch EN, Baryshnikova A, Pons C, Tan G, et al. A global genetic interaction network maps a wiring diagram of cellular function. Science (80-). 2016; <u>https://doi.org/10.1126/ science.aaf1420</u> PMID: 27708008

- Tyler AL, Donahue LR, Churchill GA, Carter GW. Weak Epistasis Generally Stabilizes Phenotypes in a Mouse Intercross. PLoS Genet. Public Library of Science; 2016; 12:e1005805. https://doi.org/10.1371/ journal.pgen.1005805 PMID: 26828925
- Varón-González C, Navarro N. Epistasis regulates the developmental stability of the mouse craniofacial shape. Heredity (Edinb). Nature Publishing Group; 2018; 1. <u>https://doi.org/10.1038/s41437-018-0140-8</u> PMID: 30209292
- Hemani G, Shakhbazov K, Westra H-J, Esko T, Henders AK, McRae AF, et al. Detection and replication of epistasis influencing transcription in humans. Nature. Nature Publishing Group; 2014; 508:249–53. https://doi.org/10.1038/nature13005 PMID: 24572353
- Park DS, Eskin I, Kang EY, Gamazon ER, Eng C, Gignoux CR, et al. An ancestry-based approach for detecting interactions. Genet Epidemiol. Wiley-Blackwell; 2018; 42:49–63. <u>https://doi.org/10.1002/gepi.</u> 22087 PMID: 29114909
- Sittig LJ, Carbonetto P, Engel KA, Krauss KS, Barrios-Camacho CM, Palmer AA. Genetic Background Limits Generalizability of Genotype-Phenotype Relationships. Neuron. Cell Press; 2016; 91:1253– 1259. https://doi.org/10.1016/j.neuron.2016.08.013 PMID: 27618673
- Lusis AJ, Seldin MM, Allayee H, Bennett BJ, Civelek M, Davis RC, et al. The Hybrid Mouse Diversity Panel: A Resource for Systems Genetics Analyses of Metabolic and Cardiovascular Traits. J Lipid Res. 2016; 58:7250–7. https://doi.org/10.1194/jlr.R066944 PMID: 27099397
- Philip VM, Sokoloff G, Ackert-Bicknell CL, Striz M, Branstetter L, Beckmann MA, et al. Genetic analysis in the Collaborative Cross breeding population. Genome Res. 2011; https://doi.org/10.1101/gr.113886. 110 PMID: 21734011
- Churchill GA, Airey DC, Allayee H, Angel JM, Attie AD, Beatty J, et al. The Collaborative Cross, a community resource for the genetic analysis of complex traits. Nature Genetics. 2004. <u>https://doi.org/10.1038/ng1104-1133 PMID: 15514660</u>
- Andreux PA, Williams EG, Koutnikova H, Houtkooper RH, Champy MF, Henry H, et al. Systems genetics of metabolism: The use of the BXD murine reference panel for multiscalar integration of traits. Cell. 2012; https://doi.org/10.1016/j.cell.2012.08.012 PMID: 22939713
- Solberg Woods LC, Palmer AA. Using Heterogeneous Stocks for Fine-Mapping Genetically Complex Traits. Methods in Molecular Biology. Humana Press Inc.; 2019. pp. 233–247. <u>https://doi.org/10.1007/</u> 978-1-4939-9581-3\_11 PMID: 31228160
- King EG, Merkes CM, McNeil CL, Hoofer SR, Sen S, Broman KW, et al. Genetic dissection of a model complex trait using the Drosophila Synthetic Population Resource. Genome Res. 2012; <u>https://doi.org/ 10.1101/gr.134031.111</u> PMID: 22496517
- Kang HM, Zaitlen NA, Wade CM, Kirby A, Heckerman D, Daly MJ, et al. Efficient control of population structure in model organism association mapping. Genetics. 2008; 178:1709–1723. <u>https://doi.org/10. 1534/genetics.107.080101</u> PMID: 18385116
- Gonzales NM, Seo J, Hernandez Cordero AI, St. Pierre CL, Gregory JS, Distler MG, et al. Genome wide association analysis in a mouse advanced intercross line. Nat Commun. 2018; <u>https://doi.org/10. 1038/s41467-018-07642-8 PMID: 30514929</u>
- Bloom JS, Boocock J, Treusch S, Sadhu MJ, Day L, Oates-Barker H, et al. Rare variants contribute disproportionately to quantitative trait variation in yeast. Elife. 2019; <u>https://doi.org/10.7554/eLife.49212</u> PMID: 31647408
- Jha SK, Rauniyar K, Jeltsch M. Key molecules in lymphatic development, function, and identification. Ann Anat—Anat Anzeiger. 2018; 219:25–34. <u>https://doi.org/10.1016/j.aanat.2018.05.003</u> PMID: 29842991
- Afratis NA, Selman M, Pardo A, Sagi I. Emerging insights into the role of matrix metalloproteases as therapeutic targets in fibrosis. Matrix Biol. 2018; 68–69:167–179. <u>https://doi.org/10.1016/j.matbio.2018</u>. 02.007 PMID: 29428229
- Cerdà-Costa N, Gomis-Rüth FX. Architecture and function of metallopeptidase catalytic domains. Protein Sci. 2014; 23:123–44. Available: http://www.ncbi.nlm.nih.gov/pubmed/24596965 https://doi.org/10. 1002/pro.2400 PMID: 24596965
- Bennett BJ, Farber CR, Orozco L, Kang HM, Ghazalpour A, Siemers N, et al. A high-resolution association mapping panel for the dissection of complex traits in mice. Genome Res. 2010; 20:281–290. https://doi.org/10.1101/gr.099234.109 PMID: 20054062
- Sul JH, Bilow M, Yang W-Y, Kostem E, Furlotte N, He D, et al. Accounting for Population Structure in Gene-by-Environment Interactions in Genome-Wide Association Studies Using Mixed Models. Schork NJ, editor. PLOS Genet. 2016; 12:e1005849. <u>https://doi.org/10.1371/journal.pgen.1005849</u> PMID: 26943367

- Ghazalpour A, Rau CDCD, Farber CRCR, Bennett BJBJ, Orozco LDLD, Van Nas A, et al. Hybrid mouse diversity panel: a panel of inbred mouse strains suitable for analysis of complex genetic traits. Mamm Genome. 2012; 23:680–92. https://doi.org/10.1007/s00335-012-9411-5 PMID: 22892838
- Rau CD, Wang J, Avetisyan R, Romay MC, Martin L, Ren S, et al. Mapping genetic contributions to cardiac pathology induced by beta-adrenergic stimulation in mice. Circ Cardiovasc Genet. 2015; 8. https:// doi.org/10.1161/CIRCGENETICS.113.000732 PMID: 25480693
- Parks BWW, Sallam T, Mehrabian M, Psychogios N, Hui STT, Norheim F, et al. Genetic Architecture of Insulin Resistance in the Mouse. Cell Metab. 2015; 21:334–346. https://doi.org/10.1016/j.cmet.2015. 01.002 PMID: 25651185
- Rau CD, Wang J, Avetisyan R, Romay MC, Martin L, Ren S, et al. Mapping genetic contributions to cardiac pathology induced by beta-adrenergic stimulation in mice. Circ Cardiovasc Genet. 2015; 8:40–49. https://doi.org/10.1161/CIRCGENETICS.113.000732 PMID: 25480693
- Wang JJ-C, Rau C, Avetisyan R, Ren S, Romay MCMC, Stolin G, et al. Genetic Dissection of Cardiac Remodeling in an Isoproterenol-Induced Heart Failure Mouse Model. PLoS Genet. 2016; 12:1–30. https://doi.org/10.1371/journal.pgen.1006038 PMID: 27385019
- Gibbs GM, Scanlon MJ, Swarbrick J, Curtis S, Gallant E, Dulhunty AF, et al. The cysteine-rich secretory protein domain of Tpx-1 is related to ion channel toxins and regulates ryanodine receptor Ca2+ signaling. J Biol Chem. American Society for Biochemistry and Molecular Biology; 2006; 281:4156–63. https://doi.org/10.1074/jbc.M506849200 PMID: 16339766
- Zhang S, Liu X, Bawa-Khalfe T, Lu L-S, Lyu YL, Liu LF, et al. Identification of the molecular basis of doxorubicin-induced cardiotoxicity. Nat Med. 2012; 18:1639–1642. https://doi.org/10.1038/nm.2919 PMID: 23104132
- Albino-Sanchez M, Vazquez-Hernandez J, Ocadiz-Delgado R, Serafin-Higuera N, León-Galicia I, Garcia-Villa E, et al. Decreased RARβ expression induces abundant inflammation and cervical precancerous lesions. Exp Cell Res. 2016; 346:40–52. https://doi.org/10.1016/j.yexcr.2016.05.010 PMID: 27207583
- Wyler DJ. Fibrosin, a Novel Fibrogenic Protein: Discovery, Cloning and Implications for Fibrotic Disorders. Int Arch Allergy Immunol. 1996; 111:326–329. <u>https://doi.org/10.1159/000237388</u> PMID: 8957104
- 35. White GR, Varley JM, Heighway J. Genomic structure and expression profile of LPHH1, a 7TM gene variably expressed in breast cancer cell lines. Biochim Biophys Acta. 2000; 1491:75–92. Available: http://www.ncbi.nlm.nih.gov/pubmed/10760572 https://doi.org/10.1016/s0167-4781(00)00020-8 PMID: 10760572
- Rau CD, Romay MC, Tuteryan M, Wang JJ-C, Santolini M, Ren S, et al. Systems Genetics Approach Identifies Gene Pathways and Adamts2 as Drivers of Isoproterenol-Induced Cardiac Hypertrophy and Cardiomyopathy in Mice. Cell Syst. Elsevier Inc.; 2017; 4:121–128.e4. <u>https://doi.org/10.1016/j.cels.</u> 2016.10.016 PMID: 27866946
- Lara-Pezzi E, Gómez-Salinero J, Gatto A, García-Pavía P. The Alternative Heart: Impact of Alternative Splicing in Heart Disease. J Cardiovasc Transl Res. 2013; 6:945–955. <u>https://doi.org/10.1007/s12265-013-9482-z</u> PMID: 23775418
- Schumer M, Brandvain Y. Determining epistatic selection in admixed populations. Mol Ecol. 2016; https://doi.org/10.1111/mec.13641 PMID: 27061282
- Ehrenreich IM. Epistasis: Searching for interacting genetic variants using crosses. Genetics. 2017. https://doi.org/10.1534/genetics.117.203059 PMID: 28592494
- Corbett-Detig RB, Zhou J, Clark AG, Hartl DL, Ayroles JF. Genetic incompatibilities are widespread within species. Nature. 2013; https://doi.org/10.1038/nature12678 PMID: 24196712
- Srivastava A, Morgan AP, Najarian ML, Sarsani VK, Sigmon JS, Shorter JR, et al. Genomes of the mouse collaborative cross. Genetics. 2017; https://doi.org/10.1534/genetics.116.198838 PMID: 28592495
- 42. Wang Y, Alla V, Goody D, Gupta SK, Spitschak A, Wolkenhauer O, et al. Epigenetic factor EPC1 is a master regulator of DNA damage response by interacting with E2F1 to silence death and activate metastasis-related gene signatures. Nucleic Acids Res. 2016; 44:117–133. https://doi.org/10.1093/nar/gkv885 PMID: 26350215
- Kee HJ, Kim J-R, Nam K-I, Park HY, Shin S, Kim JC, et al. Enhancer of Polycomb1, a Novel Homeodomain Only Protein-binding Partner, Induces Skeletal Muscle Differentiation. J Biol Chem. 2007; 282:7700–7709. https://doi.org/10.1074/jbc.M611198200 PMID: 17192267
- Combarros O, Cortina-Borja M, Smith AD, Lehmann DJ. Epistasis in sporadic Alzheimer's disease. Neurobiol Aging. 2009; 30:1333–1349. <u>https://doi.org/10.1016/j.neurobiolaging.2007.11.027</u> PMID: 18206267

- 45. Badano JL, Leitch CC, Ansley SJ, May-Simera H, Lawson S, Lewis RA, et al. Dissection of epistasis in oligogenic Bardet–Biedl syndrome. Nature. Nature Publishing Group; 2006; 439:326–330. <u>https://doi.org/10.1038/nature04370 PMID: 16327777</u>
- 46. Mackay TF, Moore JH. Why epistasis is important for tackling complex human disease genetics. Genome Med. BioMed Central; 2014; 6:125. https://doi.org/10.1186/gm561 PMID: 25031624
- Buchner DA, Nadeau JH. Contrasting genetic architectures in different mouse reference populations used for studying complex traits. Genome Res. Cold Spring Harbor Laboratory Press; 2015; 25:775– 91. https://doi.org/10.1101/gr.187450.114 PMID: 25953951
- Ackerman KG, Huang H, Grasemann H, Puma C, Singer JB, Hill AE, et al. Interacting genetic loci cause airway hyperresponsiveness. Physiol Genomics. 2005; 21:105–111. https://doi.org/10.1152/ physiolgenomics.00267.2004 PMID: 15657107
- Street VA, Kujawa SG, Manichaikul A, Broman KW, Kallman JC, Shilling DJ, et al. Resistance to Noise-Induced Hearing Loss in 129S6 and MOLF Mice: Identification of Independent, Overlapping, and Interacting Chromosomal Regions. J Assoc Res Otolaryngol. Springer US; 2014; 15:721–738. <u>https://doi.org/10.1007/s10162-014-0472-x PMID: 24952082</u>
- 50. Fuchs SBA, Lieder I, Stelzer G, Mazor Y, Buzhor E, Kaplan S, et al. GeneAnalytics: An Integrative Gene Set Analysis Tool for Next Generation Sequencing, RNAseq and Microarray Data. Omi A J Integr Biol. 2016; https://doi.org/10.1089/omi.2015.0168 PMID: 26983021